Mauna Loa Solar Observatory Observer's Log ______ Sun Oct 2 17:22:07 GMT 2011 Year: 11 Doy: 275 Observer: koon WEATHER COMMENT: Sun Oct 2 17:22:09 GMT 2011 Clear sky, wind=10 mph from the SSE, temp=50F. end Sun Oct 2 17:30:30 GMT 2011 MKTV Start Patrol Sun Oct 02 17:30:22 GMT 2011 H-ALPHA Start Patrol Sun Oct 2 17:30:59 GMT 2011 CHIP Start Patrol Sun Oct 02 17:35:33 GMT 2011 COMP Start Patrol Sun Oct 2 17:51:16 GMT 2011: PSPT Start Patrol Sun Oct 2 18:01:40 GMT 2011 CHIP Sun Oct 2 18:03:12 GMT 2011 CHIP End LSD Sun Oct 2 18:03:30 GMT 2011 CHIP BiasLSD Sun Oct 2 18:04:20 GMT 2011 CHIP End BiasLSD Sun Oct 2 18:04:28 GMT 2011 CHIP Bias Sun Oct 2 18:05:05 GMT 2011 CHIP End Bias Sun Oct 2 18:05:12 GMT 2011 CHIP ReStart Patrol **PSPT PROBLEM COMMENT BY DARRYL**: Sun Oct 2 18:21:46 GMT 2011 Crashed, restarted. The "ftr home" command was running on the PC and not stopping, and I could hear a tone coming from the instrument, after doing a "KillPspt" I pressed the Enter button on the PC to stop the "ftr home" command and then I ran "Daily Obs" to continue, so far it seems to be doing well. end Sun Oct 2 18:22:08 GMT 2011: PSPT Start Patrol Sun Oct 2 18:25:01 GMT 2011: PSPT Start Patrol WEATHER COMMENT: Sun Oct 2 19:30:48 GMT 2011 Cirrostratus has moved in, pausing coronal instruments. end Sun Oct 02 19:32:45 GMT 2011 COMP End Patrol Sun Oct 2 19:35:02 GMT 2011 MKIV End Patrol ****EVENT COMMENT BY DARRYL****: Sun Oct 2 19:37:42 GMT 2011 Lots of flaring in the NW active regions. end NOTE BY DARRYL: Sun Oct 2 19:43:00 GMT 2011 Tried gluing the guider telescope cover with silicone sealer this time. end NOTE BY DARRYL: Sun Oct 2 20:37:50 GMT 2011 Adjusted main dome slot. end GONG COMMENT BY DARRYL: Sun Oct 2 22:06:35 GMT 2011 I put the shipment of spare HDD in the GONG shelter, on floor next to blue cabinet on optical bench side.

____end_

```
Sun Oct 2 23:02:42 GMT 2011
                              CHIP
                                      LSD
Sun Oct 2 23:04:09 GMT 2011
                              CHIP
                                      End LSD
Sun Oct 2 23:04:19 GMT 2011
                              CHIP
                                      BiasLSD
Sun Oct 2 23:04:53 GMT 2011
                                      End BiasLSD
                              CHIP
Sun Oct 2 23:05:01 GMT 2011
                                      Bias
                              CHIP
Sun Oct 2 23:05:25 GMT 2011
                                      End Bias
                              CHIP
Sun Oct 2 23:05:34 GMT 2011
                              CHIP
                                      ReStart Patrol
**PSPT PROBLEM COMMENT BY DARRYL**: Sun Oct 2 23:37:22 GMT 2011
I fixed the dome rotator control box that was reported as partially pulled off
the wall, I added a better wall anchor.
____end___
Sun Oct 2 23:59:48 GMT 2011: PSPT Start Patrol
Mon Oct 3 00:17:48 GMT 2011
                              CHIP
                                      End Patrol
WEATHER COMMENT: Mon Oct 3 00:17:53 GMT 2011
Clouds too thick for spar tracking, shutting down instruments.
end
Mon Oct 3 00:18:34 GMT 2011: PSPT Abort Patrol
Mon Oct 03 00:20:17 GMT 2011 H-ALPHA End Patrol
Mon Oct 3 01:11:51 GMT 2011
       MkIV
17_30.rawmk4 17_57.rawmk4 18_23.rawmk4 18_49.rawmk4 19_20.rawmk4
17_33.rawmk4 17_59.rawmk4 18_26.rawmk4 18_52.rawmk4 19_23.rawmk4
17_36.rawmk4 18_02.rawmk4 18_29.rawmk4 18_55.rawmk4 19_25.rawmk4
17_39.rawmk4 18_05.rawmk4 18_32.rawmk4 18_58.rawmk4 19_28.rawmk4
17_42.rawmk4 18_08.rawmk4 18_35.rawmk4 19_01.rawmk4 19_31.rawmk4
17 45.rawmk4 18 11.rawmk4 18 38.rawmk4 19 04.rawmk4
17 48.rawmk4 18 14.rawmk4 18 41.rawmk4 19 10.rawmk4
```